Author's response to reviews

Title: Simple technique for the evacuation of traumatic subcutaneous haematomas under tension.

Authors:

George MR. Chami (georgechami@hotmail.com)
Edward DR. Hatley (edwardhatley@yahoo.co.uk)
Belinda DR. Chami (belindachami@hotmail.com)
Hossam MR. Dabis (h.dabis@btopenworld.com)

Version: 2 Date: 1 September 2005

Author’s response to reviews:

Dear Sir or Madam,

we would like to submit a new manuscript, please note the following revisions have been made to the previous manuscript following the reviewer's report:
1- The abstract has been constructed using 4 headings: background, methods, results and conclusions.
2- The references has been numbered consecutively and cited according to the journal style.
3- Authors' contributions and competing interests have been added to the manuscript.
4- Figures labels and legends have been corrected.
5- The wrong use of term or language errors has been corrected.
6- One Author, who made a recent significant contribution to the manuscript, has been added.

we believe that this simple technique is effective, cost effective and could be a limb saving procedure in certain circumstance when other technique are either not available or not possible, in the accident and emergency setting, as shown in the scenarios presented in the manuscript.

Kind regards

George Chami
Clinical Research Fellow, Orthopaedics.